Author's response to reviews

Title: Spontaneous Preterm Labor is Associated with an Increase in the Proinflammatory Signal Transducer TLR4 Receptor on Maternal Blood Monocytes.

Authors:

Edyta Pawelczyk (pawelczyke@cc.nih.gov)
Bogdan J Nowicki (bnowicki@mmc.edu)
Michael G Izban (mizban@mmc.edu)
Pratap Siddharth (spratap@mmc.edu)
Sashti A. Nupur (Nupur.A.Sashti@tn.gov)
Sanderson Maureen (ssanderson@mmc.edu)
Stella Nowicki (snowicki@mmc.edu)

Version: 2 Date: 4 October 2010

Author's response to reviews: see over
Subject: 6008630413165693 - Spontaneous Preterm Labor is ...

Dear Editor,

Enclosed please find a resubmission of our manuscript entitled "Spontaneous Preterm Labor is Associated with an Increase in the Proinflammatory Signal Transducer TLR4 Receptor on Maternal Blood Monocytes."

Thank you so much for the tentative acceptance of our manuscript. We addressed the editorial comments and modified our manuscript as suggested.

1. We ask that you add a "Competing Interests" section to your manuscript.

We included a 'Competing interests' section between the Conclusions and Authors' contributions. There are none "Competing Interests" to declare; The statement 'The authors declare that they have no competing interests' is now included in the text.

Thanks again

Bogdan Nowicki MD, PhD,